Author's response to reviews

Title: Biomechanical comparison of a novel transoral atlantoaxial anchored cage with established fixation technique - a finite element analysis

Authors:

Baocheng Zhang (baocheng255@163.com)
Hai-bo Liu (liuhaibo718905@126.com)
Xian-hua Cai (xianhua_cai@126.com)
Zhi-hua Wang (xianhua_cai@126.com)
Feng Xu (gkxf79390@sohu.com)
Hui Kang (kanghuicss@163.com)
Ran Ding (dana-doctor@hotmail.com)
Xiao-qing Luo (qingqingchong@126.com)

Version: 2 Date: 21 March 2015

Author's response to reviews:

Dear editor:

Thanks for your guide.

We have changed the manuscript to address here. 1. Add the line and page numbers of the manuscript. 2. Ethics committee approval for use of individual participant data was granted by the ethics committee of Wuhan General Hospital prior to this study. 3. Informed consent in the study was also obtained from the participant. 4. We added one author Xiao-qing Luo, who have edited the manuscript and wanted to use mathematical method to analyze the results.

Thanks.

Baocheng